Author's response to reviews

Title: Effect of commercial breakfast fibre cereals compared with corn flakes on postprandial blood glucose, gastric emptying and satiety in healthy subjects: a randomized blinded crossover trial

Authors:

Joanna Hlebowicz MD (joanna.hlebowicz@skane.se)
Jennie Wickenberg Student (jennie.wickenberg.647@student.lu.se)
Rickard Fahlstrom Student (lak02rfm@student.lu.se)
Ola Bjorgell PhD (Ola.Bjorgell@med.lu.se)
Lars-Olof Almer PhD (Lars-Olof.Almer@med.lu.se)
Gassan Darwiche PhD (Gassan.Darwiche@skane.se)

Version: 3 Date: 9 August 2007

Author's response to reviews:

Dear Dr Kumagai, 2007.08.09 Malmo

I would like to thank you for your important and valuable comments regarding the content of the present manuscript. We have clearly been able to reply to all your questions and remarks, as follows below. I realize that the revised present form of the manuscript will improve the presentation of our work. Do not hesitate to contact me further if you have any additional questions. I am hoping for further cooperation in the future.

Please notice that the changes in the abstract have been done as required.
1. The conclusion in the abstract has been re-orginazed.
2. Statically has been removed in the results in the abstract, line 43.

Sincerely,
Joanna Hlebowicz, MD

Dept. of Medicine, Malmo University Hospital
Ingang 35, 205 02 Malmo, Sweden
Telephone nr +46-40-331000
Fax nr +46-40-336208
E-mail: Joanna.Hlebowicz@med.lu.se